Author's response to reviews

Title: The influences of origin on beginning medical students' perceptions of rural medical practice.

Authors:

Robin A Ray (robin.ray@jcu.edu.au)
Louise Young (louise.young1@jcu.edu.au)
Daniel Lindsay (daniel.lindsay1@jcu.edu.au)

Version: 2 Date: 11 April 2014

Author's response to reviews: see over
Dear Professor Marques

Uploaded to your journal site, please find our revised paper titled: *The influences of origin on beginning medical students’ perceptions of rural medical practice.*

We apologise for omitting the Acknowledgements section. We have completed the section with the following information:

*We acknowledge the work of Dr Judy Taylor who provided peer review for the project. The Faculty of Medicine, Health and Molecular Science at James Cook University provided a small grant used to pay a research assistant. The Faculty did not contribute in any way to the design or management of the project.*

We commend our paper to you and ask you to consider this work for publication in *BMC Medical Education*

Yours sincerely

Robin Ray
Corresponding author